%% BioMed_Central_Tex_Template_v1.06
%%                                      %
%  bmc_article.tex            ver: 1.06 %
%                                       %

%%IMPORTANT: do not delete the first line of this template
%%It must be present to enable the BMC Submission system to
%%recognise this template!!

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                     %%
%%  LaTeX template for BioMed Central  %%
%%     journal article submissions     %%
%%                                     %%
%%          <8 June 2012>              %%
%%                                     %%
%%                                     %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                                                 %%
%% For instructions on how to fill out this Tex template           %%
%% document please refer to Readme.html and the instructions for   %%
%% authors page on the biomed central website                      %%
%% http://www.biomedcentral.com/info/authors/                      %%
%%                                                                 %%
%% Please do not use \input{...} to include other tex files.       %%
%% Submit your LaTeX manuscript as one .tex document.              %%
%%                                                                 %%
%% All additional figures and files should be attached             %%
%% separately and not embedded in the \TeX\ document itself.       %%
%%                                                                 %%
%% BioMed Central currently use the MikTex distribution of         %%
%% TeX for Windows) of TeX and LaTeX.  This is available from      %%
%% http://www.miktex.org                                           %%
%%                                                                 %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%% additional documentclass options:
%  [doublespacing]
%  [linenumbers]   - put the line numbers on margins

%%% loading packages, author definitions

%\documentclass[twocolumn]{bmcart}% uncomment this for twocolumn layout and comment line below
\documentclass[doublespacing,linenumbers,A4]{bmcart}

%%% Load packages
%\usepackage{amsthm,amsmath}
%\RequirePackage{natbib}
%\RequirePackage[authoryear]{natbib}% uncomment this for author-year bibliography
%\RequirePackage{hyperref}
\usepackage[utf8]{inputenc} %unicode support
%\usepackage[applemac]{inputenc} %applemac support if unicode package fails
%\usepackage[latin1]{inputenc} %UNIX support if unicode package fails


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                             %%
%%  If you wish to display your graphics for   %%
%%  your own use using includegraphic or       %%
%%  includegraphics, then comment out the      %%
%%  following two lines of code.               %%
%%  NB: These line *must* be included when     %%
%%  submitting to BMC.                         %%
%%  All figure files must be submitted as      %%
%%  separate graphics through the BMC          %%
%%  submission process, not included in the    %%
%%  submitted article.                         %%
%%                                             %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%\def\includegraphic{}
\def\includegraphics{}



%%% Put your definitions there:
\startlocaldefs
\usepackage{graphicx}
\endlocaldefs


%%% Begin ...
\begin{document}
\begin{thebibliography}{1}

\bibitem{stratrandom}
Kernan WN, Viscoli CM, Makuch RW, Brass LM, Horwitz RI.
\newblock Stratified Randomization for Clinical Trials.
\newblock Journal of Clinical Epidemiology. 1999;52:19--26.

\bibitem{AtkinsonDonev}
Atkinson AC, Donev AN, Tobias RD.
\newblock Optimum Experimental Designs, with SAS.
\newblock Oxford: Oxford University Press; 2007.

\bibitem{CookNachtsheim}
Cook RD, Nachtsheim CJ.
\newblock Computer-aided blocking for factorial and response-surface designs.
\newblock Technometrics. 1989;31:339--346.

\bibitem{Schoen_JQT}
Schoen ED.
\newblock Optimum designs versus orthogonal arrays for main effects and
  two-factor interactions.
\newblock Journal of Quality Technology. 2010;42:197--208.

\bibitem{phenflexref}
Wopereis S, Stroeve JHM, Stafleu A, Bakker GCM, Burggraaf J, Van~Erk MJ, et~al.
\newblock Multi-parameter comparison of a standardized mixed meal tolerance
  test in healthy and type 2 diabetic subjects: the PhenFlex challenge.
\newblock Genes and Nutrition. 2017;12:21.

\bibitem{minsufbal}
Sajobi TT, Singh G, Lowerison MW, Engbers J, Menon BK, Demchuk AM, et~al.
\newblock Minimal sufficient balance randomization for sequential randomized
  controlled trial designs: results from the ESCAPE trial.
\newblock Trials. 2017;18:516.

\end{thebibliography}


%%% Start of article front matter
\begin{frontmatter}

\begin{fmbox}
\dochead{Research}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% Enter the title of your article here     %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\title{Randomization of subjects to study-arms of a parallel study in the presence of multiple covariates}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% Enter the authors here                   %%
%%                                          %%
%% Specify information, if available,       %%
%% in the form:                             %%
%%   <key>={<id1>,<id2>}                    %%
%%   <key>=                                 %%
%% Comment or delete the keys which are     %%
%% not used. Repeat \author command as much %%
%% as required.                             %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\author[
   addressref={aff1,aff2},                   % id's of addresses, e.g. {aff1,aff2}
%   corref={aff1},                       % id of corresponding address, if any
   noteref={n1},                        % id's of article notes, if any
   email={eric.schoen@kuleuven.be}           % email address
]{\inits{ED}\fnm{Eric D.} \snm{Schoen}}
\author[
   addressref={aff1},
   noteref={n1},
   email={suzan.wopereis@tno.nl}
]{\inits{S}\fnm{Suzan} \snm{Wopereis}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% Enter the authors' addresses here        %%
%%                                          %%
%% Repeat \address commands as much as      %%
%% required.                                %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\address[id=aff1]{%                           % unique id
  \orgname{TNO}, % university, etc
  \street{Utrechtseweg 48},                     %
  \postcode{3700 AJ}                                % post or zip code
  \city{Zeist},                              % city
  \cny{Netherlands}                                    % country
}
\address[id=aff2]{%
  \orgname{Faculty of Bioengineering Sciences, KU Leuven},
  \street{Kasteelpark Arenberg 30—Box 2456},
  \postcode{3001}
  \city{Leuven},
  \cny{Belgium}
}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% Enter short notes here                   %%
%%                                          %%
%% Short notes will be after addresses      %%
%% on first page.                           %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{artnotes}
%\note{Sample of title note}     % note to the article
\note[id=n1]{Equal contributor} % note, connected to author
\end{artnotes}

\end{fmbox}% comment this for two column layout

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% The Abstract begins here                 %%
%%                                          %%
%% Please refer to the Instructions for     %%
%% authors on http://www.biomedcentral.com  %%
%% and include the section headings         %%
%% accordingly for your article type.       %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\begin{abstractbox}

\begin{abstract} % abstract
\parttitle{Background}
Parallel intervention studies involving healthy volunteers usually require a procedure to allocate the subjects to study-arms. Statistical models to evaluate the different outcomes of the study-arms will include study-arm as a factor along with any covariate that might affect the results. To ensure that the effects of the covariates are confounded to the least possible extent with the effects of the arms, stratified randomization can be applied. However, there is at present no clear-cut procedure when there are multiple covariates.
\parttitle{Methods}
We propose a D-optimal blocking procedure to allocate subjects with known values of the covariates to the study arms. We prove that the procedure minimizes the variances of the baseline differences between the arms corrected for the covariates. The procedure uses standard statistical software.
\parttitle{Results}
We demonstrate the potential of the method by an application to a human parallel intervention trial with three arms and 162 healthy volunteers. The covariates were gender, age, body mass index, an initial composite health score, and a categorical indicator called first-visit group, defining groups of volunteers who visit the clinical centre on the same day (17 groups). Volunteers were allocated equally to the study-arms by the D-optimal blocking procedure. The D-efficiency of the model connecting an outcome with the study-arms and correcting for the covariates equals 99.2\%. We simulated 10,000 random allocations of subjects to arms either unstratified or stratified by first-visit group. Intervals covering the middle 95\% of the D-efficiencies for these allocations were [82.0, 92.0] and [93.2, 98.4], respectively.
\parttitle{Conclusions}
Allocation of volunteers to study-arms with a D-optimal blocking procedure with the values of the covariates as inputs substantially improves the efficiency of the statistical model that connects the response with the study arms and corrects for the covariates.
\parttitle{Trial registration}
Dutch Trial Register NL7054 (NTR7259). Registered May 15, 2018, https://www.trialregister.nl/trial/7054.
\end{abstract}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                                          %%
%% The keywords begin here                  %%
%%                                          %%
%% Put each keyword in separate \kwd{}.     %%
%%                                          %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{keyword}
\kwd{D-efficiency}
\kwd{Randomization}
\kwd{Blocking}
\end{keyword}

% MSC classifications codes, if any
%\begin{keyword}[class=AMS]
%\kwd[Primary ]{}
%\kwd{}
%\kwd[; secondary ]{}
%\end{keyword}

%\end{abstractbox}
%
%\end{fmbox}% uncomment this for twocolumn layout

\end{frontmatter}



\section*{Background}
A clinical trial involving volunteers usually focuses on comparing the effect of different interventions or treatments of the subjects. The main instrument for the comparison is a statistical model that links clinical outcomes to the intervention or treatment that was applied. An important concern when evaluating the treatment differences is the possibility of bias in the treatment effects. For example, if the subjects allocated to one of the arms in a parallel or between-subjects study design are on average older than those allocated to another arm, the crude difference in average responses of these arms may be biased by age.

There are well-known methods to obtain unbiased estimates of the treatment effects. First, for biases due to subject-specific covariates, one could consider using a cross-over or within-subjects design, in which any subject receives all of the treatments, separated by a wash-out period. However, the treatments may be incompatible with such a design. For example, lifestyle interventions such as weight reduction studies should involve a between-subjects rather than a within-subjects design, because effects are not easily washed out.

For parallel or between-subjects studies, each subject is allocated to one of the arms. Bias due to covariates can be reduced in two ways. First, the randomization procedure to allocate the subjects to the arms could be made to address specific sources of bias. For example, one could use a stratified randomization \cite{stratrandom} to reduce bias due to age. This involves defining age classes, determining the age class for each subject and randomizing the allocation of the subjects of each age class to the arms of the study.  As a consequence, the arms of the study have about the same average value of the covariate involved so that differences in treatment outcome means hardly need adjustment for differences in a covariate among the arms. Stratified randomization is a study design-based measure to reduce bias.

The second way to reduce bias in between-subjects studies is to include the potential sources of bias in the statistical model to evaluate study outcomes. The statistical model then provides estimates of the treatment effects corrected for covariates such as age. Inclusion of the covariate in the statistical model in addition to stratified randomization hardly removes bias, but it does reduce the unexplained variation within the arms. Therefore, the effects of the different treatments are estimated more precisely than without the inclusion of the covariate in the model.

It might be possible to define homogeneous study groups based on multiple covariates. When these groups are sufficiently large, stratified randomization could proceed in the same way as for a single covariate. However, with an increasing number of covariates the groups become too small for a stratified randomization. Nevertheless, it is desirable that the arms of a human study have about the same average value for all of the covariates, or, in case of a categorical covariate, shows a proportional distribution of that covariate for the respective arms. The motivating example for this paper is a human intervention study with three study-arms, in which gender, age, boy mass index (BMI), and a composite health score are continuous covariates. Subsequently, a categorical covariate called first-visit group defines groups of subjects whose initial visit of the clinical center was on the same day. There are 17 different first-visit groups so that a stratified randomization addressing all the covariates would be impossible to perform.

The purpose of this paper is to propose a method to allocate subjects to the arms of a study using multiple covariates. The method is model-based such that the variances of the intervention effects corrected for the covariate are minimized. It can be conducted using commercially available statistical software, requiring as input a list specifying for each subject the covariates that need to be addressed, and the number of arms in the study. The output is an allocation table where each subject is allocated to one of the arms. We further propose a quantitative measure of the effectiveness of any allocation. We show  the effectiveness of the proposed allocation method for the motivating example using both this measure and tables of mean values or frequencies of the covariates for the three study-arms. We compare the effectiveness of our procedure with results on 10,000 completely random allocations and 10,000 stratified random allocations within each of the 17 levels of the first-visit group. Finally, we sum up benefits and possible drawbacks of the procedure.


\section*{Methods}
\subsection*{Allocation procedure}
Our method to allocate subjects to the arms of a study is based on a statistical model that relates any parameter of interest to the arms of the study and the covariates considered to be relevant for that study. We assume that these parameters are continuous so that we can use linear-model methodology. Let $y$ be the $N \times 1$ vector of observed values of these parameters. Further, let $X$ be the $N \times p$ matrix of covariates and $T$ the $N \times (t-1)$ matrix of contrasts modelling the differences between the $t$ arms of the study. The statistical model of the data is then

\begin{equation}\label{eq:satmodel}
y=X\beta+T \gamma+\epsilon,
\end{equation}

where $\beta$ is a $p \times 1$ vector of coefficients corresponding to the covariates, $\gamma$ is a $(t-1) \times 1$ vector of coefficients quantifying the differences between the arms, and $\epsilon ~ N(0,\sigma^2 I_N)$ is a vector of identically and independently distributed normal variables with mean 0 and variance $\sigma^2$, and $I_N$ is the $N \times N$ identity matrix.

The purpose of the study is to estimate $\gamma$ as precisely as possible. \cite{AtkinsonDonev} show that the precision of the estimator of $\gamma$ is maximized if

\begin{equation}\label{eq:Dsoptimality}
\mathcal{D}_s=|T^T (I-X(X^T X)^{-1} X^T )T|
\end{equation}

is maximized, where $|.|$ denotes the determinant. This is determinant of the residual sums of squares and products matrix after regressing the contrast vectors making up the differences between the $t-1$ arms of the study effects in $T$ on the covariates collected in $X$. Denoting the full matrix of parameters with $F= [X T]$, it can be shown that (\ref{eq:Dsoptimality}) is equivalent to

\begin{equation}\label{eq:Dssimple}
\mathcal{D}_s=|F^T F|/|X^T X|
\end{equation}

The matrix $X$ is fixed when allocating the subjects to the arms of the study. The only way to affect $D$  is in the allocation of the subjects to the arms as expressed with the matrix $T$. We propose optimizing this allocation using the blocking procedure of \cite{CookNachtsheim}. This procedure groups the rows of a pre-existing matrix $T$ with treatment columns into groups, or blocks, of a specified size such that the precision of the estimator of $\gamma$ is maximized. Denoting the blocking allocation matrix by $X$ and using $F = [T X]$, the procedure maximizes

\begin{equation}\label{eq:Dsreverse}
D^*=|F^TF|/|T^TT|.
\end{equation}

The blocking factor is a categorical covariate added by the procedure. In the present case, however, we have a pre-existing matrix of covariates X and we want to allocate the rows to groups corresponding to the arms. If we fix the number of subjects in each arm, we can nevertheless use the blocking procedure to obtain an optimal allocation. Fixing the number of subject per arm does not fix the matrix $T$, but it does fix $|T^T T|$. Therefore, $|F^T F|$ is maximized by the procedure. By \ref{eq:Dsreverse}, maximization of $|F^T F|$ implies maximization of $D$ so that the procedure can be used to optimize the allocation of subjects to the arms of a study.

\subsection*{Measuring effectiveness}
By writing the determinant in (\ref{eq:Dsoptimality}) as
\begin{equation}\label{eq:Dsalternative}
\mathcal{D}_s=|T^TT - T^TX(X^T X)^{-1} X^T T|,
\end{equation}
it can be seen that the maximum is reached if $X$ and $T$ are orthogonal to each other, and $D_s=|T^T|$. Without loss of generality, $T$ can be made to include $t-1$ orthogonal contrasts with length $\sqrt{N}$. Therefore, $|T^TT| = N^{t-1}$. Following \cite{Schoen_JQT}, we define the $D_s$ efficiency of the study design with respect to the covariates as $\mathcal{D}_s^{1/(t-1)}/N$. Its value equals 1 if the allocation is such that the treatment contrasts are orthogonal to the covariates. Its value equals 0 if one or both of the treatment contrasts are completely aliased with one of the covariates.


%the covariates to allocate subjects to study-arms are determined prior to the actual treatment. In cross-over designs there are
%Many parallel studies have assessments of the subjects at several time points, and the interest is in determining whether the change over time depends on the study-arm. There is a similarity with cross-over studies, because there are at least two measurements over time. A statistical model for this case should include random effects to model differences between subjects and the interaction \verb/study-arm x time/ to asses whether the change over time is the same in each of the study-arms. A correction for covariates based on only their main effects would affect the measurements at the time points in the same amount. As a result, changes of study outcomes during the study are unaffected by these main effects. Therefore, it hardly makes sense to include the main effects of these covariates in the statistical model if that model also accounts for differences between subjects. However, it does make sense to include both the covariates' main effects and interaction terms of the form \verb/covariate x time/ in addition to the interaction \verb/study-arm x time/. Alternatively, one can consider the change over time as the primary outcome and formulate a statistical model include both the study-arm and the main effects of the covariates. We conclude that the proposed randomization procedure is also applicable for parallel study designs with subject evaluations at several time points.


\section*{Results}
\subsection*{Performance of the allocation procedure}
The motivating example for this paper is a clinical study to assess the effects of general nutritional advice versus personalized nutritional advice to control a set of health parameters in volunteers. There were three study arms: (1) a control arm where no advice was offered, (2) a general nutritional advice arm and (3) a personalized nutritional advice arm. The study included a total of 162 subjects. Each of the subjects visited the clinical center three times. The initial visit involved collecting baseline information of the subjects such as BMI. Subjects furthermore consumed a mixed meal challenge test drink and multiple blood samples were obtained afterwards to analyse the physiological response. Based on this response, an initial health score was calculated \cite{phenflexref}. The intervention started on the second visit to the clinical center. The main interest of the study is in the relation between the study arm and changes in health score between second and third visit.

Gender, age, BMI and initial health score may affect the change in health parameters  between the second and third visit. Therefore, they should be included as covariates in a statistical model relating the change in health score between second and third visit to the arm of the study. We decided also to include the grouping variable that indicates which participants initially visited the clinical center on the same day. There are 17 first-visit groups indicated by this variable. These groups of subjects remain largely the same for the second and third visits. Therefore, including the first-visit group as a variable in a statistical model accounts for different changes in health parameters for people starting early in the study when compared to those starting later on, for example by seasonal variation.

In view of the main interest of the study, the differences between the arms as regards the changes in health parameters between the second and third visit should be confounded to the least possible extent with differences between gender, age, BMI, initial health score and first-visit group. The matrix $X$ in equation~\ref{eq:satmodel} thus includes one column for each of the parameters gender, age, BMI and initial health score and 17 columns indicating whether a subject belongs to first-visit group 1 up to 17 (an entry 1 for subject $i$ in column $j$ indicates that subject $i$ belongs to first-visit group $j$).The matrix $T$ is a $162 \times 2$ matrix of normalized contrasts making up the differences between the 3 arms of the study. An allocation of subjects to the arms of the study corresponds to a permutation in the rows of $T$ against a fixed matrix $X$. We impose the restriction that each arm is to include 54 subjects. Therefore, $|T^TT|$ in equation~\ref{eq:Dsreverse} equals $162^2$.

We used SAS/QC procedure OPTEX with 5000 iterations to allocate the subjects to the arms. The $D_s$-efficiency of the allocation is 0.992. Table~\ref{tab:arms_gender} shows the distribution over the arms for female (F) and male (M) subjects. The table shows that the 104 female subjects are allocated as evenly as possible to the three arms. The same is the case for the 58 male subjects. Neither 104 nor 58 is divisible by 3 so that the distribution cannot be perfectly even. The distribution is such that each arm includes 54 subjects and the ratio of female over male is as close to constant as possible.
Table~\ref{tab:arms_firstvisit} shows the distribution over the arms for each first-visit group. Note that these groups are of unequal size. It is clear that the subjects of one and the same first-visit group are allocated as evenly as possible to the three arms. Finally, the means and standard deviations of the continuous covariates age, BMI and initial health score are shown in Table~\ref{tab:arms_continuous}. The three arms have nearly the same mean values of these covariates, while their standard deviations are also close. We conclude that the randomization procedure was very successful in returning an efficient study design.

\section*{Discussion}
\subsection*{Comparison with complete and restricted randomization}
To assess the amount of improvement due to the proposed procedure, we performed 10,000 random allocations of the subjects to arms and evaluated for each allocation the $D_s$ efficiency for the study design with respect to the covariates. We considered a completely random allocation and a random allocation stratified by first-visit group. For the latter case, we assumed that, for each first-visit group, the numbers of subjects assigned to each of the arms correspond to the numbers in Table~\ref{tab:arms_firstvisit}.

The results of the random allocation are shown in Figure~1. The blue bars show a frequency distribution of the 10,000 random allocations. The proposed procedure, addresses all covariates simultaneously. The small red bar denotes the $D_s$-efficiency of 99.2 obtained in this way. Intervals covering 95\% of the $D_s$-efficiencies for the completely random and stratified random allocations were [0.80, 0.90] and [0.93, 0.98], respectively. Maximum values were 0.97 and 0.99 respectively. We conclude that it is very likely that a random allocation returns a substantially worse $D_s$-efficiency for the study design than a model-based allocation. Even in case the numbers of subjects per arm are predetermined for each first-visit group, the stratified randomization generally produces inferior study designs.

\subsection*{Restriction to specific study designs}
The allocation procedure proposed here is appropriate for a parallel study design in which the subjects are simultaneously allocated to the arms. This implies that the procedure is not compatible with studies where an immediate choice among treatments is required such as clinical trials that enrol subjects gradually. Sajobi et al. \cite{minsufbal} provide a heuristic method to balance multiple covariates as much as is feasible over the arms of the study. The method presented in that paper employs an updated table of the current distribution of each individual covariate over the arms to allocate a new subject to one of these arms. Instead, our method operates directly on the precision of the parameter estimators modeling the differences between the arms when corrected for the joint effects of the covariates.

The procedure is also not compatible with a cross-over design, because such a design features several treatments per subject. Therefore, a correction for the covariates determined prior to the actual start of the study affects these treatments in the same amount. As a result, treatment differences in cross-over studies are unaffected by covariates measured at the start.

\section*{Conclusions}
In this paper, we proposed a model-based procedure to allocate subjects with known values of a set of covariates to the study-arms of a parallel study. The procedure can be viewed as an extension of stratified randomization to the case of multiple covariates. It operates using standard statistical software. Its goal function is such that the allocation results in a maximum precision for the effects of the study-arms after correction for the covariates. The success of the procedure can be measured by the $D_s$-efficiency  of the resulting design and by summary tables classified by the covariate values and the study-arm. An application to a human intervention trial with three arms and 162 volunteers showed that the procedure is highly effective.

\begin{backmatter}
\section*{Abbreviations}
BMI: Body Mass Index.

\section*{Declarations}
\subsection*{Ethics approval and consent to participate}
Before entering the study, all participants provided written informed consent. The study was approved by the French medical Ethics Committee, Comité de Protection des Personnes (CPP) and agreed upon by the French health agency (ANSM, Competent Regulatory Authority). The study was conducted in accordance with the Declaration of Helsinki as revised in 1983 and was registered in the Dutch Trial Register NL7054 (NTR7259) on May 15, 2018; see https://www.trialregister.nl/trial/7054.

\subsection*{Consent for publication}
Not applicable.

\subsection*{Availability of data and materials}
Data used for the allocation are available upon reasonable request. SAS code for the allocation is provided as a supplementary file.

\subsection*{Competing interests}
The authors declare that they have no competing interests.

\subsection*{Funding}
The work was supported by the public private partnership entitled “PhenFlex-based resilience as measure for health effects of diet” financed by Topsector Agri \& Food (TKI-AF-16035), and co-funded by the PPP Allowance made available by Health~Holland, Top Sector Life Sciences \& Health to stimulate public-private partnerships. Additional funds were provided by TNO roadmap Biomedical Health, Pfizer Inc, BASF SE, By-Health, Roquette, Biofortis and CIRO. The funders of the study had no role in the study design, data interpretation, or the preparation of the manuscript.

\subsection*{Author's contributions}
EDS and SW developed the study design; EDS developed the randomization procedure; SW supervised the study; EDS and SW wrote the paper; both authors read and approved the final paper.

\subsection*{Acknowledgements}
We gratefully acknowledge Biofortis for recruiting the subjects and conducting the study. This paper was approved by the PhenFlex2 consortium.

\bibliographystyle{vancouver} % Style BST file (bmc-mathphys, vancouver, spbasic).
%\bibliography{bmc_ref}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%                               %%
%% Figures                       %%
%%                               %%
%% NB: this is for captions and  %%
%% Titles. All graphics must be  %%
%% submitted separately and NOT  %%
%% included in the Tex document  %%
%%                               %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%
%% Do not use \listoffigures as most will included as separate files

\section*{Figure ledends}
  \begin{figure}[!h]
  \caption{\csentence{D$_s$-efficiencies for potential complete randomizations in the motivating example.}
  Based on a model with gender, age, BMI, a composite health score and membership of one of 17 first-visit groups as covariates, efficiencies of the estimators of the effects of 3 study arms are shown for 10,000 random allocations of subject to arms, not stratified by any covariate. The red bar denotes the efficiency obtained with the proposed procedure, which addresses all covariates simultaneously.}
   %   \includegraphics[width=12cm]{fig1.eps}
      \end{figure}

  \begin{figure}[!h]
  \caption{\csentence{D$_s$-efficiencies for potential stratified randomizations in the motivating example.}
  Based on a model with gender, age, BMI, a composite health score and membership of one of 17 first-visit groups as covariates, efficiencies of the estimators of the effects of 3 study arms are shown for 10,000 random allocations of subject to arms, stratified by first-visit group. The red bar denotes the efficiency obtained with the proposed procedure, which addresses all covariates simultaneously.}
      % \includegraphics[width=12cm]{fig2.eps}
      \end{figure}

\section*{Tables}
\begin{table}[h!]
\caption{Allocation of subjects to study arms according to gender in the motivating example. The table shows the numbers of female (F) and male (M) subjects allocated to the arms A, B and C, respectively.}\label{tab:arms_gender}
      \begin{tabular}{cccc}
      \hline
Gender	& 	\multicolumn{3}{c}{Arm}\\
\hline
	&	A	&	B	&	C	\\
F	&	34	&	35	&	35	\\
M	&	20	&	19	&	19	\\
\hline
      \end{tabular}
\end{table}

\begin{table}[h!]
\caption{Allocation of subjects to study-arms according to first-visit group in the motivating example. A first-visit group comprises the subjects that had their intake visit on the same day. The table shows for each first-visit group the numbers of subjects allocated to the arms A, B and C, respectively.}\label{tab:arms_firstvisit}
      \begin{tabular}{cccc}
\hline		
Visit	&		\multicolumn{3}{c}{Arm}					\\
	&		A	&	B	&	C	\\
\hline
1	&		3	&	4	&	3	\\
2	&		3	&	3	&	4	\\
3	&		4	&	4	&	3	\\
4	&		3	&	2	&	2	\\
5	&		4	&	3	&	4	\\
6	&		3	&	3	&	4	\\
7	&		3	&	2	&	2	\\
8	&		4	&	4	&	3	\\
9	&		2	&	3	&	3	\\
10	&		3	&	3	&	3	\\
11	&		3	&	3	&	3	\\
12	&		3	&	4	&	4	\\
13	&		3	&	3	&	4	\\
14	&		4	&	3	&	3	\\
15	&		4	&	4	&	4	\\
16	&		2	&	2	&	2	\\
17	&		3	&	4	&	3	\\
\hline
      \end{tabular}
\end{table}

\begin{table}[h!]
\caption{Allocation of subjects to study-arms according to age, BMI and initial health score in the motivating example. The table shows means and standard deviations of the covariates in the arms A, B and C, respectively.}\label{tab:arms_continuous}
      \begin{tabular}{cccccccccc}
      \hline
      Covariate &&\multicolumn{8}{c}{Arm}\\
	&&	\multicolumn{2}{c}{A}		&&	\multicolumn{2}{c}{B	}		&&	\multicolumn{2}{c}{C}\\
	&&	mean	&	sd	&&	mean	&	sd	&&	mean	&	sd	\\
\hline
Age	&&	42.63	&	12.59	&&	42.33	&	12.36	&&	42.93	&	11.87	\\
BMI	&&	25.28	&	3.15	&&	25.31	&	3.68	&&	25.27	&	3.97	\\
Health Score	&&	1.476	&	0.135	&&	1.475	&	0.121	&&	1.478	&	0.133	\\
\hline
      \end{tabular}
\end{table}

\clearpage
\section*{Additional Files}
  \subsection*{Randomization.txt}
SAS code to allocate the subjects to the three arms of the motivating example study.

\end{backmatter}
\end{document}
